Author's response to reviews

Title: Association of IL1A and IL1B loci with primary open angle glaucoma

Authors:

Suddhasil Mookherjee (suddhasilmookherjee@gmail.com)
Deblina Banerjee (banerjee.deblina@gmail.com)
Subhadip Chakraborty (subhadipiicb@gmail.com)
Antara Banerjee (banerjee.antara77@gmail.com)
Indranil Mukhopadhyay (indranilm100@gmail.com)
Abhijit Sen (skeya063@gmail.com)
Kunal Ray (kunalray@gmail.com)

Version: 10 Date: 18 June 2010

Author's response to reviews: see over
Dear Dr. Edmunds,

We have completed the proofreading of the manuscript entitled “Association of IL1A and IL1B loci with primary open angle glaucoma”. The document is checked for typographical errors and is fixed.

We hope that the editors will find our manuscript suitable for publication in BMC Medical Genetics.

Sincerely,

Kunal Ray